Author's response to reviews

Title: Acute and chronic antihypertensive effects of Cinnamomum zeylanicum stem bark methanol extract in L-NAME-induced hypertensive rats

Authors:

Paulin Nyadjeu (epauka@yahoo.fr)
Elvine Pami Nguelefack-Mbuyo (mbuyopamielvine@gmail.com)
Albert Donatien Atsamo (atsamoalbert@yahoo.fr)
Telesphore Benoît Nguelefack (nguelefack@yahoo.fr)
Alain Dongmo (alainberd@yahoo.fr)
Albert Kamanyi (Akamanyi@hotmail.com)

Version: 4 Date: 30 September 2012

Author's response to reviews: see over
To

The Editor BMC Complementary and alternative Medicine

MS: 1373889339731710

“Acute and chronic antihypertensive effects of Cinnamomum zeylanicum stem bark methanol extract in L-NAME-induced hypertensive rats”

Dear Sir,
Thank you for your Email.
The changes in the pre-review revisions have been made in blue color.

The following modifications have been made:

1- Title page at the front of our manuscript file was added. This page contains, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.
2- In the Methods section, a paragraph on “Ethical consideration” (see page 5) has also been included.
3- In the method section, the paragraph entitled “Aortic and left ventricular histological analysis” has been modified by adding the method by which the aorta thickness was calculated. Thus, some information has been added to figure 5
4- Figure 2 has been modified and improved

We are waiting for the decision that will be reserved to this revised version.

Sincerely yours.